Author's response to reviews

Title: TOR complex 2 is needed for cell cycle progression and anchorage-independent growth of MCF7 and PC3 tumor cells

Authors:

Ville Hietakangas (ville@tl.t.org.sg)
Stephen M Cohen (cohen@tl.t.org.sg)

Version: 4 Date: 11 September 2008

Author's response to reviews:

Dear Ann,

Please find enclosed the modified version of our manuscript #1093612702181368. We have carefully assessed the comments of the two expert reviewers. While we agree with David Stokoe that analyzing the role of TORC2 in normal tissue and nontumorigenic cells is important, we feel that it is beyond the scope of this rather focused study. As suggested by Dr. Stokoe, we have modified the introduction of the manuscript to better reflect the specific scope of the manuscript. We hope that you will find the current version of the manuscript suitable for publication in BMC Cancer.

Best regards,

Ville Hietakangas